Author’s response to reviews

Title: The association of early-life exposure to ambient PM2.5 and later-childhood height-for-age in India: An observational study

Authors:
Dean Spears (dean@riceinstitute.org)
Sagnik Dey (sagnik@cas.iitd.ac.in)
Sourangsu Chowdhury (sourangsuchowdhury@gmail.com)
Sangita Vyas (sangita@riceinstitute.org)
Joshua Apte (jsapte@utexas.edu)

Version: 1 Date: 21 Feb 2019

Author’s response to reviews:

While this report is clearly within the topic area covered by this journal, we ask that submissions respect the author guidelines before we can undertake editorial handling.

Among major issues, Please insert a Discussion section and shorten the Conclusions to a crisp single paragraph.

-- done

We request double spacing.

-- done

Make sure that all literature cited is referenced using consecutive numbers.

-- done